Author's response to reviews

Title: Expression of pS2 in prostate cancer correlates with grade and Chromogranin A expression but not with stage

Authors:
Dr M Hammad Ather (hammad.ather@aku.edu)
Farhat Abbas (farhat.abbas@aku.edu)
Nuzhat Faruqui (nuzhat.faruqui@aku.edu)
M Israr (mohammad.israr@aku.edu)
Shahid Pervaiz (shahid.pervaiz@aku.edu)

Version: 2 Date: 29 Mar 2004

Dear Editor,

I would like to apologize for the delay in re submission of the revised manuscript.

Please note following changes have been incorporated in the manuscript based upon the recommendations of the two reviewers.

Reviewer 1 (JH)
Major compulsory revisions
1. Language; reviewed and necessary corrections made
2. Title and abstract; line 5 in methods section re written to clarify
3. Introduction; references added as advised
5. Results; last para in result section moved to second para and confusion created by pS2 confined....... is clarified in line 2 and 4-5 in second paragraph. The issue related to grade is clarified in the revised fig. 2 and para 3 in the result section
6. References; new reference added and numbering revised

Reviewer 2 (MC)
Discretionary revisions
1. introduction; second last line in the introduction deleted
Minor compulsory revision
2. Missing labels in figures added
Major compulsory revisions
3. introduction; there is evidence that NE differentiation has prognostic value, please refer to Ather MH et al Eur Urol 2000 vol 38
4. positive control; refer to para 3 line 12-15 in the methods section. All patients received hormonal treatment.

Regards,

Dr. M H Ather